Author's response to reviews

Title: Prevalence and Predictors of Resistant Hypertension in Primary Care Setting: A cross-sectional study

Authors:

YookChin Chia (chiayc@um.edu.my)
SiewMooi Ching (sm_ching@upm.edu.my)

Version: 6Date: 30 June 2014

Author's response to reviews:

Chia Yook Chin
University of Malaya Medical Centre
Department of Primary Care Medicine,
Faculty of Medicine, University of Malaya,
50603 Kuala Lumpur, Malaysia,
Phone: +603-7949 2620
Fax: +603-7957 7941
chiayc@um.edu.my

June 30, 2014
Journal Executive Editor
BMC Family Medicine
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Dear Prof Rupert Payne,

REVISED PAPER FOR PUBLICATION: Ref. MS: 3220873321101514
Title: Prevalence and predictors of Resistant Hypertension in a Primary Care Setting:
A cross-sectional study by Chia Yook Chin a *(MBBS, FRCP) Ching Siew Mooib (MD, MMed Fammed)

Thank you to you for the feedback. I removed all the tables from within the results as suggested by you and placed them under references.

I had made some changes in the title page to ensure that the revised manuscript conforms to the journal style. Below are the changes highlighted in blue which I
have made.

With warm regards,

Thank you,

Yours sincerely

Dr Chia Yook Chin
Department of Primary Care Medicine,
Faculty of Medicine,
University of Malaya,
50603 Kuala Lumpur,
MALAYSIA

Fax +603-7957 7941
Email: chiayc@um.edu.my